Author's response to reviews

Title: Short and long-term career plans of final year dental students in the United Arab Emirates

Authors:

Hazim H Rashid (hazm2001@emirates.net.ae)
Swapnil G Ghotane (swapnil.1.ghotane@kcl.ac.uk)
Salem H Abufanas (abufanas@yahoo.com)
Jennifer E Gallagher (jenny.gallagher@kcl.ac.uk)

Version: 2 Date: 17 February 2013

Author's response to reviews:

Dear Sir/Madam

The following amendments have been made to the paper and online submission as requested

1. A unique e-mail has been inserted for Dr Abufanus
2. A consent statement has been inserted into the Methods section
3. Acknowledgements including funding sources have been inserted.

We trust that it is now satisfactory to be considered acceptable to be sent for review

Yours faithfully

Jenny

Dr Jennifer E Gallagher on behalf of co-authors